Author’s response to reviews

Title: Mixed cerebrovascular disease in an elderly man with mixed vascular risk factors: a case report

Authors:

Dian He (hedian@gmc.edu.cn; ileks@163.com)

YunLi Yu (yuyunli@126.com)

Shan Wu (wuwushan@163.com)

ShuFen Tian (tianshufen919@hotmail.com)

Hui Yu (yuhui2001y@yeah.net)

Shu Xu (shu-xu1973@hotmail.com)

Lan Chu (chulan8999@126.com)

Version: 3 Date: 26 Jan 2019

Author’s response to reviews:

Dear Professor Benjamin Ragen,

We have carefully considered each comment, and made the corresponding amendments. Firstly, we made major modifications on the language and grammatical issues throughout the article. The revised expressions are much closer to the native language, and the revised manuscript is more readable. Secondly, we replaced “a case study” with “a case report” in the title, and re-wrote the background for avoiding the overlap to the reference. Thirdly, Dr. YH and Dr. XS participated in manuscript revision, especially in the current revision, their contributions have been updated. Fourthly, we added a fund, which supported the copy-editing of the manuscript, the roles of the funding were supplemented. Lastly, permissions to acknowledge were obtained from all persons mentioned in the Acknowledgements section.

We do hope this new revision will obtain your approval and acceptance. Thank you very much for your attention and consideration. Looking forward to your decision.

Sincerely,

Dian He